\begin{table}%t3
\caption{\label{Tab:err}Stellar parameters for 7 binaries with literature values.}
%\centerline
{\small\begin{tabular}{lccccc}
\hline \hline
RasTyc Name & Name & Sp.Type$^{\rm Our}$ &  Sp.Type$^{\rm Lit}$ & $T_{\rm eff}^{\rm Our}$ & $T_{\rm eff}^{\rm Lit}$  \\ \hline
RasTyc~0702$-$0515 & AR Mon & K1 III / G2 V & ... / ... & 4395 / 5350 & 4500 / 5200$^{1}$  \\
RasTyc~0921$-$0640 & NY Hya & G2 V / G2 V & ... / ... & 5450 / 5450 & 5490 / ...$^{2}$  \\
RasTyc~1310+3556 & RS Cvn & F7 V / G8 V & F6 IV / G8 IV$^{3}$ & 6120 / 5420 & ... / ...  \\
RasTyc~1601+5120 & EV Dra & G7 V / K1 V & G8 V / K1 V$^{4}$ & 5400 / 5370 & ... / ...  \\
RasTyc~1825+1817 & AW Her & G2 V / K0 IV & G2 IV / K2 IV$^{5}$  & 5780 / 4870 & ... / ...  \\
RasTyc~1912+4619 & FL Lyr & G2 V / K1 V & F8 / G8$^{6}$ & 5730 / 5080 & 6080 / 5390$^{7}$ \\ 
\hline
\end{tabular}}
\par
\medskip
References. (1) \citet{2005AJ....129.2798W}; (2) \citet{2001AetA...374..980C}; (3) \citet{1990AetA...230..389S}; (4) \citet{1995AJ....110.2926H}; (5) \citet{2002AetA...387..850I}; (6) \citet{1986AJ.....91..383P}; (7) \citet{2007ActaAstronomica...57..301F}.
\end{table}